\begin{thebibliography}{99}

\bibitem{Bona:2007qt}
  M.~Bona {\it et al.},
  %``SuperB: A High-Luminosity Asymmetric e+ e- Super Flavor Factory. Conceptual
  %Design Report,''
  arXiv:0709.0451 [hep-ex].
  %%CITATION = ARXIV:0709.0451;%%

\bibitem{Baracchini:2007ei}
  E.~Baracchini {\it et al.},
  %``Investigating the physics case of running a B-factory at the Upsilon(5S)
  %resonance,''
  JHEP {\bf 0708} (2007) 005
  [arXiv:hep-ph/0703258].
  %%CITATION = JHEPA,0708,005;%%

%% Vub part

%\cite{Aubert:2006qi}
\bibitem{Aubert:2006qi}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``Determinations of |V(ub)| from inclusive semileptonic B decays with
  %reduced model dependence,''
  Phys.\ Rev.\ Lett.\  {\bf 96} (2006) 221801
  [arXiv:hep-ex/0601046].
  %%CITATION = PRLTA,96,221801;%%
%\cite{Aubert:2006ry}
\bibitem{Aubert:2006ry}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``Measurement of the B --> pi l nu branching fraction and determination  of
  %|V(ub)| with tagged B mesons,''
  Phys.\ Rev.\ Lett.\  {\bf 97} (2006) 211801
  [arXiv:hep-ex/0607089].
  %%CITATION = PRLTA,97,211801;%%
%\cite{Abe:2006gb}
\bibitem{Abe:2006gb}
  K.~Abe {\it et al.}  [Belle Collaboration],
  %``Measurement of exclusive B --> X/u l nu decays using a full-reconstruction
  %tag at Belle,''
  arXiv:hep-ex/0610054.
  %%CITATION = HEP-EX/0610054;%%

%% bsg part
%\cite{Chen:2001fja}
\bibitem{Chen:2001fja}
  S.~Chen {\it et al.}  [CLEO Collaboration],
  %``Branching fraction and photon energy spectrum for b --> s gamma,''
  Phys.\ Rev.\ Lett.\  {\bf 87} (2001) 251807
  [arXiv:hep-ex/0108032].
  %%CITATION = PRLTA,87,251807;%%
%\cite{Abe:2001hk}
\bibitem{Abe:2001hk}
  K.~Abe {\it et al.}  [Belle Collaboration],
  %``A measurement of the branching fraction for the inclusive B --> X/s  gamma
  %decays with Belle,''
  Phys.\ Lett.\  B {\bf 511} (2001) 151
  [arXiv:hep-ex/0103042].
  %%CITATION = PHLTA,B511,151;%%
%\cite{Koppenburg:2004fz}
\bibitem{Koppenburg:2004fz}
  P.~Koppenburg {\it et al.}  [Belle Collaboration],
  %``An inclusive measurement of the photon energy spectrum in b --> s gamma
  %decays,''
  Phys.\ Rev.\ Lett.\  {\bf 93} (2004) 061803
  [arXiv:hep-ex/0403004].
  %%CITATION = PRLTA,93,061803;%%
%\cite{Aubert:2005cua}
\bibitem{Aubert:2005cua}
  B.~Aubert {\it et al.}  [BABAR Collaboration],
  %``Measurements of the $B \to X_s \gamma$ branching fraction and photon
  %spectrum from a sum of exclusive final states,''
  Phys.\ Rev.\  D {\bf 72} (2005) 052004
  [arXiv:hep-ex/0508004].
  %%CITATION = PHRVA,D72,052004;%%
%\cite{Aubert:2006gg}
\bibitem{Aubert:2006gg}
  B.~Aubert {\it et al.}  [\babar\  Collaboration],
  %``Measurement of the branching fraction and photon energy moments of B  -->
  %X/s gamma and A(CP)(B --> X(s+d) gamma),''
  Phys.\ Rev.\ Lett.\  {\bf 97} (2006) 171803
  [arXiv:hep-ex/0607071].
  %%CITATION = PRLTA,97,171803;%%
%\cite{Barberio:2006bi}
\bibitem{Barberio:2006bi}
  E.~Barberio {\it et al.}  [Heavy Flavor Averaging Group (HFAG)],
  %``Averages of b-hadron properties at the end of 2005,''
  arXiv:hep-ex/0603003.
  %%CITATION = HEP-EX/0603003;%%
%\cite{Aubert:2007my}
\bibitem{Aubert:2007my}
  B.~Aubert {\it et al.}  [BABAR Collaboration],
  %``Measurement of the B --> X_s gamma Branching Fraction and Photon Energy
  %Spectrum using the Recoil Method,''
  Phys.\ Rev.\  D {\bf 77} (2008) 051103
  [arXiv:0711.4889 [hep-ex]].
  %%CITATION = PHRVA,D77,051103;%%
\bibitem{Misiak:2006zs}
  M.~Misiak {\it et al.},
  %``The first estimate of B(anti-B --> X/s gamma) at O(alpha(s)**2),''
  Phys.\ Rev.\ Lett.\  {\bf 98} (2007) 022002
  [arXiv:hep-ph/0609232].
  %%CITATION = PRLTA,98,022002;%%
\bibitem{Becher:2006pu}
  T.~Becher and M.~Neubert,
  %``Analysis of Br(B --> X/s gamma) at NNLO with a cut on photon energy,''
  Phys.\ Rev.\ Lett.\  {\bf 98} (2007) 022003
  [arXiv:hep-ph/0610067].
  %%CITATION = PRLTA,98,022003;%%
\bibitem{misiaktalk} {\texttt http://www.theorie.physik.uni-muenchen.de/lsfritzsch/albufeira/Talks/albufeiramm.pdf}
%\cite{Lee:2006wn}
\bibitem{Lee:2006wn}
  S.~J.~Lee, M.~Neubert and G.~Paz,
  %``Enhanced non-local power corrections to the B --> X/s+ gamma decay rate,''
  Phys.\ Rev.\  D {\bf 75} (2007) 114005
  [arXiv:hep-ph/0609224].
  %%CITATION = PHRVA,D75,114005;%%

\bibitem{Haisch}
  U.~Haisch and A.~Weiler,
  %``Bound on minimal universal extra dimensions from anti-B --> X/s gamma,''
  Phys.\ Rev.\  D {\bf 76} (2007) 034014
  [arXiv:hep-ph/0703064].
  %%CITATION = PHRVA,D76,034014;%%


%%%%%%%%%%%BSLL

%\cite{Huber:2007vv}
\bibitem{Huber:2007vv}
  T.~Huber, T.~Hurth and E.~Lunghi,
  %``Logarithmically Enhanced Corrections to the Decay Rate and Forward Backward
  %Asymmetry in anti-B --> X(s) l+ l-,''
  arXiv:0712.3009 [hep-ph].
  %%CITATION = ARXIV:0712.3009;%%
%\cite{Huber:2005ig}
\bibitem{Huber:2005ig}
  T.~Huber, E.~Lunghi, M.~Misiak and D.~Wyler,
  %``Electromagnetic logarithms in anti-B --> X/s l+ l-,''
  Nucl.\ Phys.\  B {\bf 740} (2006) 105
  [arXiv:hep-ph/0512066].
  %%CITATION = NUPHA,B740,105;%%
%\cite{Ligeti:2007sn}
\bibitem{Ligeti:2007sn}
  Z.~Ligeti and F.~J.~Tackmann,
  %``Precise predictions for B -> Xs l+ l- in the large q^2 region,''
  Phys.\ Lett.\  B {\bf 653}, 404 (2007)
  [arXiv:0707.1694 [hep-ph]].
  %%CITATION = PHLTA,B653,404;%%
%\cite{Lee:2005pk}
\bibitem{Lee:2005pk}
  K.~S.~M.~Lee and I.~W.~Stewart,
  %``Shape-function effects and split matching in B --> X/s l+ l-,''
  Phys.\ Rev.\  D {\bf 74} (2006) 014005
  [arXiv:hep-ph/0511334].
  %%CITATION = PHRVA,D74,014005;%%
%\cite{Lee:2005pwa}
\bibitem{Lee:2005pwa}
  K.~S.~M.~Lee, Z.~Ligeti, I.~W.~Stewart and F.~J.~Tackmann,
  %``Universality and m(X) cut effects in B --> X/s l+ l-,''
  Phys.\ Rev.\  D {\bf 74} (2006) 011501
  [arXiv:hep-ph/0512191].
  %%CITATION = PHRVA,D74,011501;%%
%\cite{Aubert:2004it}
\bibitem{Aubert:2004it}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``Measurement of the $B \to X_s \ell^+ \ell^-$ branching fraction with a sum
  %over exclusive modes,''
  Phys.\ Rev.\ Lett.\  {\bf 93} (2004) 081802
  [arXiv:hep-ex/0404006].
  %%CITATION = PRLTA,93,081802;%%
%\cite{Iwasaki:2005sy}
\bibitem{Iwasaki:2005sy}
  M.~Iwasaki {\it et al.}  [Belle Collaboration],
  %``Improved measurement of the electroweak penguin process B --> X/s l+  l-,''
  Phys.\ Rev.\  D {\bf 72} (2005) 092005
  [arXiv:hep-ex/0503044].
  %%CITATION = PHRVA,D72,092005;%%

%% ---- Ktautau -----

%\cite{Du:1993sh}
\bibitem{Du:1993sh}
  D.~S.~Du, C.~Liu and D.~X.~Zhang,
  %``The Rare Decay B $\to$ K Tau+ Tau- In Heavy Meson Chiral Perturbation
  %Theory,''
  Phys.\ Lett.\  B {\bf 317} (1993) 179.
  %%CITATION = PHLTA,B317,179;%%
%\cite{Hewett:1995dk}
\bibitem{Hewett:1995dk}
  J.~L.~Hewett,
  %``Tau polarization asymmetry in B $\to$ X(s) tau+ tau-,''
  Phys.\ Rev.\  D {\bf 53} (1996) 4964
  [arXiv:hep-ph/9506289].
  %%CITATION = PHRVA,D53,4964;%%

%% ---- Knunu -----
%\cite{Buchalla:2000sk}
\bibitem{Buchalla:2000sk}
  G.~Buchalla, G.~Hiller and G.~Isidori,
  %``Phenomenology of non-standard Z couplings in exclusive semileptonic  b -->
  %s transitions,''
  Phys.\ Rev.\  D {\bf 63} (2001) 014015
  [arXiv:hep-ph/0006136].
  %%CITATION = PHRVA,D63,014015;%%
%\cite{Bird:2004ts}
\bibitem{Bird:2004ts}
  C.~Bird, P.~Jackson, R.~Kowalewski and M.~Pospelov,
  %``Search for dark matter in b --> s transitions with missing energy,''
  Phys.\ Rev.\ Lett.\  {\bf 93} (2004) 201803
  [arXiv:hep-ph/0401195].
  %%CITATION = PRLTA,93,201803;%%
%\cite{Georgi:2007ek}
\bibitem{Georgi:2007ek}
  H.~Georgi,
  %``Unparticle Physics,''
  Phys.\ Rev.\ Lett.\  {\bf 98} (2007) 221601
  [arXiv:hep-ph/0703260].
  %%CITATION = PRLTA,98,221601;%%
%\cite{Aliev:2007gr}
\bibitem{Aliev:2007gr}
  T.~M.~Aliev, A.~S.~Cornell and N.~Gaur,
  %``B \to K(K^*) missing energy in Unparticle physics,''
  JHEP {\bf 0707} (2007) 072
  [arXiv:0705.4542 [hep-ph]].
  %%CITATION = JHEPA,0707,072;%%
%\cite{Chen:2007zk}
\bibitem{Chen:2007zk}
  K.~F.~Chen {\it et al.}  [Belle Collaboration],
  %``Search for B -> h(*) nu nubar Decays at Belle,''
  Phys.\ Rev.\ Lett.\  {\bf 99} (2007) 221802
  [arXiv:0707.0138 [hep-ex]].
  %%CITATION = PRLTA,99,221802;%%
%\cite{Buras:2000dm}
\bibitem{Buras:2000dm}
  A.~J.~Buras, P.~Gambino, M.~Gorbahn, S.~Jager and L.~Silvestrini,
  %``Universal unitarity triangle and physics beyond the standard model,''
  Phys.\ Lett.\  B {\bf 500} (2001) 161
  [arXiv:hep-ph/0007085].
  %%CITATION = PHLTA,B500,161;%%

%% ---- lnu ----

%\cite{Ikado:2006un}
\bibitem{Ikado:2006un}
  K.~Ikado {\it et al.},
  %``Evidence of the purely leptonic decay B- --> tau- anti-nu/tau,''
  Phys.\ Rev.\ Lett.\  {\bf 97} (2006) 251802
  [arXiv:hep-ex/0604018].
  %%CITATION = PRLTA,97,251802;%%
%\cite{Aubert:2007bx}
\bibitem{Aubert:2007bx}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``A search for B+ --> tau+ nu,''
  Phys.\ Rev.\  D {\bf 76} (2007) 052002
  [arXiv:0705.1820 [hep-ex]].
  %%CITATION = PHRVA,D76,052002;%%
%\cite{Aubert:2007xj}
\bibitem{Aubert:2007xj}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``A search for B+ --> tau+ nu with Hadronic B tags,''
  Phys.\ Rev.\  D {\bf 77} (2008) 011107
  [arXiv:0708.2260 [hep-ex]].
  %%CITATION = PHRVA,D77,011107;%%
%\cite{Satoyama:2006xn}
\bibitem{Satoyama:2006xn}
  N.~Satoyama {\it et al.}  [Belle Collaboration],
  %``A search for the rare leptonic decays B+ --> mu+ nu and B+ --> e+ nu,''
  Phys.\ Lett.\  B {\bf 647} (2007) 67
  [arXiv:hep-ex/0611045].
  %%CITATION = PHLTA,B647,67;%%
%\cite{Aubert:2004yu}
\bibitem{Aubert:2004yu}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``Search for the rare leptonic decay $B^+ \to \mu^+ \nu_\mu$,''
  Phys.\ Rev.\ Lett.\  {\bf 92} (2004) 221803
  [arXiv:hep-ex/0401002].
  %%CITATION = PRLTA,92,221803;%%
%\cite{Aubert:2006at}
\bibitem{Aubert:2006at}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``A search for the decays B+ --> e+ nu/e and B+ --> mu+ nu/mu using
  %hadronic-tag reconstruction,''
  arXiv:hep-ex/0607110.
  %%CITATION = HEP-EX/0607110;%%
%\cite{Ecklund:2007zm}
\bibitem{Ecklund:2007zm}
  K.M.~Ecklund {\it et al.}  [CLEO Collaboration],
  %``Measurement of the Absolute Branching Fraction of D_s^+ --> tau^+ nu_tau
  %Decay,''
  arXiv:0712.1175 [hep-ex].
  %%CITATION = ARXIV:0712.1175;%%

%% Emi Kou

%\cite{DescotesGenon:2002ja}
\bibitem{DescotesGenon:2002ja}
  S.~Descotes-Genon and C.~T.~Sachrajda,
  %``Universality of nonperturbative QCD effects in radiative B decays. ((U)),''
  Phys.\ Lett.\  B {\bf 557} (2003) 213
  [arXiv:hep-ph/0212162].
  %%CITATION = PHLTA,B557,213;%%
%\cite{Korchemsky:1999qb}
\bibitem{Korchemsky:1999qb}
  G.~P.~Korchemsky, D.~Pirjol and T.~M.~Yan,
  %``Radiative leptonic decays of B mesons in QCD,''
  Phys.\ Rev.\  D {\bf 61} (2000) 114510
  [arXiv:hep-ph/9911427].
  %%CITATION = PHRVA,D61,114510;%%
%\cite{Aubert:2007yh}
\bibitem{Aubert:2007yh}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``Search for the radiative leptonic decay B+ --> gamma l+ nu,''
  arXiv:0704.1478 [hep-ex].
  %%CITATION = ARXIV:0704.1478;%%
%\cite{Aubert:2006ft}
\bibitem{Aubert:2006ft}
  B.~Aubert {\it et al.}  [\babar\  Collaboration],
  %``A search for the B0 --> e+ e- gamma and B0 --> mu+ mu- gamma decays,''
  arXiv:hep-ex/0607058.
  %%CITATION = HEP-EX/0607058;%%
%\cite{Yilmaz:2004tr}
\bibitem{Yilmaz:2004tr}
  U.~O.~Yilmaz, B.~B.~Sirvanli and G.~Turan,
  %``A general analysis of the lepton polarizations in B/s --> gamma l+ l-
  %decays,''
  Nucl.\ Phys.\  B {\bf 692} (2004) 249
  [arXiv:hep-ph/0407006].
  %%CITATION = NUPHA,B692,249;%%
%\cite{Choudhury:2005rz}
\bibitem{Choudhury:2005rz}
  S.~R.~Choudhury, A.~S.~Cornell, N.~Gaur and G.~C.~Joshi,
  %``Signatures of new physics in dileptonic B-decays,''
  Int.\ J.\ Mod.\ Phys.\  A {\bf 21} (2006) 2617
  [arXiv:hep-ph/0504193].
  %%CITATION = IMPAE,A21,2617;%%
%\cite{Hiller:2004wc}
\bibitem{Hiller:2004wc}
  G.~Hiller and A.~S.~Safir,
  %``Predictions for B --> K gamma gamma decays,''
  JHEP {\bf 0502} (2005) 011
  [arXiv:hep-ph/0411344].
  %%CITATION = JHEPA,0502,011;%%
%\cite{Bertolini:1998hp}
\bibitem{Bertolini:1998hp}
  S.~Bertolini and J.~Matias,
  %``The b --> s gamma gamma transition in softly broken supersymmetry,''
  Phys.\ Rev.\  D {\bf 57} (1998) 4197
  [arXiv:hep-ph/9709330].
  %%CITATION = PHRVA,D57,4197;%%
%\cite{Aubert:2001fm}
\bibitem{Aubert:2001fm}
  B.~Aubert {\it et al.}  [\babar\ Collaboration],
  %``Search for the decay $B^0 \to \gamma \gamma$,''
  Phys.\ Rev.\ Lett.\  {\bf 87} (2001) 241803
  [arXiv:hep-ex/0107068].
  %%CITATION = PRLTA,87,241803;%%
%\cite{Abe:2005bs}
\bibitem{Abe:2005bs}
  K.~Abe {\it et al.}  [Belle Collaboration],
  %``Search for the decay B0 --> gamma gamma,''
  Phys.\ Rev.\  D {\bf 73} (2006) 051107
  [arXiv:hep-ex/0507036].
  %%CITATION = PHRVA,D73,051107;%%

%% -- comparisons
%\cite{Browder:2007gg}
\bibitem{Browder:2007gg}
  T.~Browder, M.~Ciuchini, T.~Gershon, M.~Hazumi, T.~Hurth, Y.~Okada and A.~Stocchi,
  %``On the Physics Case of a Super Flavour Factory,''
  JHEP {\bf 0802} (2008) 110
  [arXiv:0710.3799 [hep-ph]].
  %%CITATION = JHEPA,0802,110;%%
%\cite{Browder:2008em}
\bibitem{Browder:2008em}
  T.~E.~Browder, T.~Gershon, D.~Pirjol, A.~Soni and J.~Zupan,
  %``New Physics at a Super Flavor Factory,''
  arXiv:0802.3201 [hep-ph].
  %%CITATION = ARXIV:0802.3201;%%


%% ---- pheno ----

\bibitem{Allanach:2002nj}
  B.~C.~Allanach {\it et al.},
  %``The Snowmass points and slopes: Benchmarks for SUSY searches,''
  % in {\it Proc. of the APS/DPF/DPB Summer Study on the Future of Particle Physics (Snowmass 2001) } ed. N.~Graf,
  % {\it In the Proceedings of APS / DPF / DPB Summer Study on the Future of Particle Physics (Snowmass 2001), Snowmass, Colorado, 30 Jun - 21 Jul
  % 2001, pp P125}
  arXiv:hep-ph/0202233.
  %%CITATION = ECONF,C01063;%%

\bibitem{Weiglein:2004hn}
  G.~Weiglein {\it et al.}  [LHC/LC Study Group],
  %``Physics interplay of the LHC and the ILC,''
  Phys.\ Rept.\  {\bf 426} (2006) 47
  [arXiv:hep-ph/0410364].
  %%CITATION = PRPLC,426,47;%%

\bibitem{Borjanovic:2005tv}
   I.~Borjanovic, J.~Krstic and D.~Popovic,
   %``SUSY studies with Snowmass Point 5 mSUGRA parameters,''
   Czech.\ J.\ Phys.\  {\bf 55} (2005) B793.
   %%CITATION = CZYPA,55,B793;%%

\bibitem{Buchmueller:2007zk}
  O.~Buchmueller {\it et al.},
  %``Prediction for the Lightest Higgs Boson Mass in the CMSSM using Indirect
  %Experimental Constraints,''
  Phys.\ Lett.\  B {\bf 657} (2007) 87
  [arXiv:0707.3447 [hep-ph]].
  %%CITATION = PHLTA,B657,87;%%

\bibitem{Goto:2007ee}
  T.~Goto, Y.~Okada, T.~Shindou and M.~Tanaka,
  %``Patterns of flavor signals in supersymmetric models,''
  arXiv:0711.2935 [hep-ph].
  %%CITATION = ARXIV:0711.2935;%%

%% ---- interplay ----

%\cite{delAguila:2008iz}
\bibitem{delAguila:2008iz}
  F.~del Aguila {\it et al.},
  %``Collider aspects of flavor physics at high Q,''
  arXiv:0801.1800 [hep-ph].
  %%CITATION = ARXIV:0801.1800;%%
%\cite{Buchalla:2008jp}
\bibitem{Buchalla:2008jp}
  G.~Buchalla {\it et al.},
  %``B, D and K decays,''
  arXiv:0801.1833 [hep-ph].
  %%CITATION = ARXIV:0801.1833;%%
\bibitem{Raidal:2008jk}
  M.~Raidal {\it et al.},
  %``Flavour physics of leptons and dipole moments,''
  arXiv:0801.1826 [hep-ph].
  %%CITATION = ARXIV:0801.1826;%%
%\cite{Hurth:2003th}
\bibitem{Hurth:2003th}
  T.~Hurth and W.~Porod,
  %``Non-diagonal flavor observables in B and collider physics,''
  Eur.\ Phys.\ J.\  C {\bf 33} (2004) S764
  [arXiv:hep-ph/0311075].
  %%CITATION = EPHJA,C33,S764;%%
\bibitem{interplay1} {\texttt https://twiki.cern.ch/\\
twiki/bin/view/Main/ColliderAndFlavour}
\bibitem{interplay2} {\texttt http://indico.cern.ch/\\
conferenceOtherViews.py?view=standard\&confId=22180}

\end{thebibliography}
